Author’s response to reviews

Title: Barriers and facilitators to uptake of cervical cancer screening among women in Uganda: a systematic review

Authors:
Eleanor Black (ellie.black@gmail.com)
Fran Hyslop (f.hyslop@unsw.edu.au)
Robyn Richmond (R.Richmond@unsw.edu.au)

Version: 3 Date: 30 Jul 2019

Author’s response to reviews:

July 2019

Professor Manuel Zorzi
Editor, BMC Women’s Health


Eleanor Black, MIPH, MBBS/BA.; Fran Hyslop, BSc; Robyn Richmond, DSc, PhD, MHEd, MA

Dear Professor Zorzi

Thank you for inviting us to revise our paper and re-submit to BMC Women’s Health. We have amended our manuscript in accordance with all the editor’s comment and are pleased to submit a revised version of our paper. Specifically, we have

• Amended the author’s contributions section to detail the contributions of all authors
• Ensured that all figures and tables are cited within the text
• Moved Tables 1 and 3 to the end of the manuscript
• Removed unnecessary files
• Uploaded a single, final, clean version

All authors have contributed to and approved this submitted revision of the paper. We look forward to hearing from you regarding publication of our paper.

Fran Hyslop
School of Public Health and Community Medicine Faculty of Medicine,
University of New South Wales,
Kensington, NSW, 2052, AUSTRALIA. CRICOS Provider Code 00098G
Email: f.hyslop@unsw.edu.au
Tel: +61 (2) 9385 1746 Fax: +61 (2) 9313 6185